Author's response to reviews

Title: OK-DIABETES: Supported self-management for adults with type 2 diabetes and a learning disability: Study protocol a randomized controlled feasibility trial

Authors:

Rebecca EA Walwyn (R.E.A.Walwyn@leeds.ac.uk)
Amy M Russell (a.m.russell@leeds.ac.uk)
Louise D Bryant (l.d.bryant@leeds.ac.uk)
Amanda J Farrin (a.j.farrin@leeds.ac.uk)
Alex Wright-Hughes (a.wright-hughes@leeds.ac.uk)
Elizabeth H Graham (e.h.graham@leeds.ac.uk)
Claire T Hulme (c.t.hulme@leeds.ac.uk)
John L O’Dwyer (j.odwyer@leeds.ac.uk)
Gary Latchford (g.latchford@leeds.ac.uk)
Alison Stansfield (alisonstansfield@nhs.uk)
Dinesh Nagi (Dinesh.Nagi@midyorks.nhs.uk)
Ramzi Ajjan (r.ajjan@leeds.ac.uk)
Allan O House (a.o.house@leeds.ac.uk)

Version: 2
Date: 20 March 2015

Author's response to reviews:

I have made the formatting changes requested by the editorial team prior to review.